Author’s response to reviews

Title: Arthroscopic rotator cuff repair with and without subacromial decompression is safe and effective

Authors:

Umile Giuseppe Longo (g.longo@unicampus.it;longoug@gmail.com)
Stefano Petrillo (s.petrillo@unicampus.it)
Vincenzo Candela (v.candela@unicampus.it)
Giacomo Rizzello (g.rizzello@unicampus.it)
Mattia Loppini (m.loppini@unicampus.it)
Nicola Maffulli (maffulli@unicampus.it)
Vincenzo Denaro (denaro.cbm@gmail.com)

Version: 4  Date: 24 Dec 2019

Author’s response to reviews:

BMSD-D-19-00100R3
RE: “Arthroscopic rotator cuff repair with and without subacromial decompression is safe and effective.”

1. Overlap

We note that the current submission contains some textual overlap with other previously published works, in particular:


While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

We thank the AE for his suggestion. We modified the manuscript and reformulated the overlapping text.
2. Acknowledgements
Please remove the 'All authors have read and approved the manuscript' from after your Acknowledgements as this is already included in the Authors’ Contributions section.

We thank the AE for his suggestion. Modified as required.

3. Poster
We note you have published your abstract in a poster previously:
DOI 10.1007/s00167-014-2917-5

Please acknowledge this in your Acknowledgements and cite it appropriately.

We thank the AE for his suggestion. We rewrite the abstract.

We hope that the changes implemented have improved the manuscript, and that this has now reached the standard necessary for formal acceptance.
We look forward to hearing from you.